Author’s response to reviews

Title: Risk factors and prognosis of young stroke. The FUTURE study: a prospective cohort study. Study rationale and protocol

Authors:

Loes CA Rutten-Jacobs (l.jacobs@neuro.umcn.nl)
Noortje AM Maaijwee (n.maaijwee@neuro.umcn.nl)
Renate M Arntz (r.arntz@neuro.umcn.nl)
Mayte van Alebeek (m.vanalebeek@neuro.umcn.nl)
Pauline Schaapsmeersd (p.schaapsmeers@mps.umcn.nl)
Henny C Schoonderwaldt (h.schoonderwaldt@neuro.umcn.nl)
Lucille DA Dorresteijn (l.dorresteijn@mst.nl)
Sebastiaan Overeem (s.overeem@neuro.umcn.nl)
Gea Droste (g.drost@neuro.umcn.nl)
Mirian C Janssen (m.janssen@aig.umcn.nl)
Waander L van Heerde (w.vanheerde@chl.umcn.nl)
Roy PC Kessels (r.kessels@donders.ru.nl)
Marcel P Zwiers (marcel.zwiers@fcdonders.ru.nl)
David G Norris (d.norris@fcdonders.ru.nl)
Maureen J van der Vlugt (m.vandervlugt@cardio.umcn.nl)
Ewoud J van Dijk (e.vandijk@neuro.umcn.nl)
Frank-Erik de Leeuw (h.deleeuw@neuro.umcn.nl)

Version: 4 Date: 28 August 2011

Author’s response to reviews: see over
Dear Prof Alam,

Thank you for giving us the opportunity to submit a new version of our manuscript, entitled ‘Risk factors and prognosis of young stroke. The FUTURE study: a prospective cohort study. Study rationale and protocol’, with your suggested revision. We changed the numbers and sequence of the two tables, so now the two tables are cited in numerical order.

We hope we sufficiently corrected your request and are looking forward to its publication as a ‘Study Protocol’ in BMC Neurology.

Yours sincerely,

Noortje AM Maaijwee, MD, on behalf of Loes CA Rutten-Jacobs, MSc
Frank Erik de Leeuw, MD, PhD, Corresponding author
Radboud University Nijmegen Medical Centre, Donders Institute for Brain, Cognition and Behaviour
Department of Neurology (326)
PO Box 9101, 6500 HB, Nijmegen
The Netherlands
0031-24-3613394 (tel.)/ 0031-24-3541122 (fax)
N.Maaijwee@neuro.umcn.nl
L.Jacobs@neuro.umcn.nl
H.deLeeuw@neuro.umcn.nl